Author's response to reviews

Title: The MDRD equation underestimates the prevalence of CKD among blacks and overestimates the prevalence of CKD among whites compared to the CKD-EPI equation: A retrospective cohort study.

Authors:

Pradeep Arora (Pradeep.Arora@va.gov)
Srini Rajagopalan (srini_rajagopalan@yahoo.com)
Nilang Patel (nilang_patel@yahoo.com)
Neha Nainani (nnainani@buffalo.edu)
Rocco C Venuto (rvenuto@ecmc.edu)
James W Lohr (James.Lohr@va.gov)

Version: 6 Date: 10 January 2012

Author's response to reviews: see over
January 10, 2012

Jigisha Patel, MRCP, Ph.D.
Series Editor, Medicine

Dear Dr. Patel:

We are pleased to have our manuscript accepted for publication in BMC Nephrology. We have carefully reviewed the manuscript and made the formatting changes as requested. We have also cropped the figures as per request. Please let us know if further changes are needed.

Thank you for your consideration of this revised manuscript.

Best regards,

Pradeep Arora, MD
Nilang Patel, MD
James Lohr, MD